Author's response to reviews

Title: Comparison of Single and Two-Tunnel Techniques during Open Treatment of Acromioclavicular Joint Disruption: a Retrospective study

Authors:

Zhiyong Hou (drzyhou@gmail.com)
Jove Graham (jhgraham1@geisinger.edu)
Kent Strohecker (kastrohecker@geisinger.edu)
Yingze Zhang (dryzzhang@126.com)
Daniel Feldmann (ddfelfmann@geisinger.edu)
Thomas R Bowen (trbowen@geisinger.edu)
Wei Chen (surgeonchenwei@126.com)
Wade R Smith (wadesmith2@gmail.com)

Version: 6 Date: 15 July 2014

Author's response to reviews:

Dear Editor Nonato:

Thank you very much for your comments. I have revised the manuscript following your comments, and the changed made in the manuscript is highlightened. The reply to the comments is listed as follows:

Comments and replies:
1. Title page: Please remove the author suffixes from the title page
Reply: The author suffixes have been removed from the title page.

2. Ethics and Consent
Thank you for including details of waiver for patient consent in your manuscript. However, we would ask you to please include the full name of the committee that provided ethical approval for your study in this statement.
Reply: The full name of the committee is Institutional Review Board of Geisinger Medical Center, which has been included in the last sentence of Method section.

3. Abbreviations
Please provide a list of abbreviations used in the text:
Reply: The "Abbreviations" have been provided.

4. Acknowledgements
Please provide an Acknowledgements section in your manuscript:
http://www.biomedcentral.com/bmcsurg/authors/instructions/researcharticle#formatting-acknowledgements
Reply: The "Acknowledgements" section has been provided.
5. Figures

Thank you for providing figure legends for your images. Unfortunately there are still some adjustments to be made. You have provided 4 images but refer to the figures as 1 A and B and 2A and B. Please either amend your figure legends (and the relevant text) to reflect 4 figures or provide 2 composite illustration files which contain all parts of the figures: http://www.biomedcentral.com/bmcsurg/authors/instructions/researcharticle#preparing-figures

Reply: There are four images in the manuscript, and they be referred to figure 1, 2, 3, 4, respectively. And the legends has also been revised accordingly.

Please contact me if there is any more question regarding the manuscript.

With best regards,

Sincerely yours,

Zhiyong Hou